

Effectiveness and safety of dupilumab in the treatment of atopic dermatitis in children (6–11 years): data from a French multicentre retrospective cohort in daily practice

Audrey Lasek, Nathalia Bellon, Sylvain Mallet, Ève Puzenat, Anne-Claire Bursztejn, Claire Abasq, Juliette Mazereeuw-Hautier, Christine Chiaverini, Thomas Hubiche, Nadia Raison-peyron, et al.

▶ To cite this version:

Audrey Lasek, Nathalia Bellon, Sylvain Mallet, Ève Puzenat, Anne-Claire Bursztejn, et al.. Effectiveness and safety of dupilumab in the treatment of atopic dermatitis in children (6–11 years): data from a French multicentre retrospective cohort in daily practice. Journal of the European Academy of Dermatology and Venereology, 2022, 36 (12), pp.2423 - 2429. 10.1111/jdv.18450. hal-03903398

HAL Id: hal-03903398

https://univ-rennes.hal.science/hal-03903398v1

Submitted on 16 Dec 2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Accepted Artic

Title: Effectiveness and safety of dupilumab in the treatment of atopic dermatitis in children

(6-11 years): data from a French multicenter retrospective cohort in daily practice

Keywords: atopic dermatitis; dupilumab; children; daily practice

Manuscript word count: 2419 words

Abstract word count: 268

Figures: 2

Tables: 5

A Lasek¹, N Bellon², S Mallet³, E Puzenat⁴, AC Bursztejn⁵, C Abasq⁶, J Mazereeuw-Hautier⁷, C Chiaverini⁸, T Hubiche⁸, N Raison Peyron⁹, A Du Thanh⁹, S Barbarot¹⁰, H Aubert¹⁰, Z Reguiai¹¹, C Droitcourt¹², C Fievet¹³, A Bellissen¹⁴, MBachelerie¹⁵, A Nosbaum¹⁶, A Leymarie¹⁷, P Armingaud¹⁸, M Masson Regnault¹⁹, E Mahé²⁰ and research Group of the Société Française de Dermatologie Pédiatrique (GR SFDP), Groupe de recherche Eczéma Atopique (GREAT)

¹ Department of dermatology, Hôpital Saint Vincent de Paul, GHICL, Lille, France

² Department of dermatology, Centre Hospitalier Universitaire Necker-Enfants Malades, Assistance Publique-Hôpitaux de Paris, Paris, France

³ Department of dermatology, venereology, and cancerology, Hôpital de la Timone, Assistancepublique-Hôpitaux de Marseille, Marseille, France

⁴ Department of dermatology, Centre Hospitalier Universitaire deBesançon, Besançon, France

⁵ Department of dermatology, Hôpitaux de Brabois, CHRU Nancy, Vandœuvre-Lès-Nancy, France

This article has been accepted for publication and undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process which may lead to differences between this version and the Version of Record. Please cite this article as doi: 10.1111/jdv.18450

- ⁶ Department of dermatology, Centre Hospitalier Universitaire de Brest, Brest, France
- ⁷ Department of dermatology, Centre Hospitalier Universitaire de Toulouse, Université Paul Sabatier Toulouse, France
- ⁸ Department of dermatology, hospital Archet 2, ESPIC CHU-Lenval, Nice, France
- ⁹ Department of dermatology, Centre Hospitalier Universitaire de Montpellier, Montpellier, France
- ¹⁰ Department of dermatology, Hôtel-Dieu, Nantes, France
- ¹¹ Department of dermatology, Polyclinique de Courlancy, Reims, France
- Department of dermatology, Centre Hospitalier Universitaire Pontchaillou, Université de Rennes, Rennes, France
- ¹³ Department of dermatology, Centre Hospitalier Régional Universitaire Lille, Lille, France
- ¹⁴ Department of dermatology, Centre Hospitalier d'Aubagne, Aubagne, France
- ¹⁵ Department of dermatology, Centre Hospitalier Universitaire de Clermond Ferrand, Clermont-Ferrand, France
- ¹⁶ Department of Clinical Immunology and Allergy, Lyon-Sud University Hospital, Pierre Bénite, Lyon, Lyon, France
- ¹⁷ Department of dermatology, Centre Hospitalier Universitaire de Caen, Caen, France
- ¹⁸ Department of dermatology, Centre Hospitalier d'Orléans, Orléans, France
- ¹⁹Department of dermatology, Centre Hospitalier de Poitiers, France
- ²⁰ Department of dermatology, Centre Hospitalier Victor Dupouy, Argenteuil, France

Corresponding author:

Accepted Articl

Audrey Lasek Duriez

Department of dermatology, Hôpital Saint Vincent de Paul, GHICL, Lille, France

lasek.audrey@ghicl.net

Funding sources: None

Conflicts of Interest:

A. Lasek has paid activities as a consultant, advisor, or speaker for Sanofi, AbbVie, and Lilly.

J.Mazereeuw-Hautier is a consultant for Sanofi and Abbvie.

E.Mahé has paid activities as a consultant, advisor, or speaker for Sanofi, AbbVie, and Janssen-

Cilag.

Other authors have no conflicts of interest to disclose.

Acknowledgement: The patients/patients' guardians in this manuscript have given written

informed consent to the publication of their case details.

Data availability statement:

This study was approved by the CNIL (Commission Nationale de l'Informatique et des Libertés).

Non-opposition to the use of their de-identified records was obtained for this non-

interventionnal study, in accordance with French legislation.

Abbreviations used:

AD: atopic dermatitis

SE: Side effect

DFR: dupilumab facial redness

IGA/ Investigator global assessment

Accepted Articl

IQR: interquartile range

SCORAD: Scoring atopic dermatitis

SCORAD50: SCORAD score improvement of at least 50%

SCORAD75: SCORAD score improvement of at least 75%

TCS: topical corticosteroids

Abstract

Background: Dupilumab is the first biotherapy available for the treatment of moderate-to-severe childhood atopic dermatitis (AD).

Objective: The aim of this study was to evaluate the effectiveness and safety of dupilumab in daily practice.

Methods: Patients aged 6 to 11, who had received a first dose of dupilumab were included in this multicenter retrospective cohort study. The primary endpoint was change in SCORAD after 3 months of treatment. Secondary endpoints were change in IGA score at 3 months, proportion of patients with SCORAD50 and SCORAD75, description of adverse events and proportion of children in our cohort who would be excluded from pivotal phase 3 clinical trial.

Results: 80 patients were included. After 3 months of treatment, there was a significant decrease in SCORAD (mean: $21.8 \pm 13.8 \text{ vs} 53.9 \pm 18.5$; p<0.0001) and IGA ($1.3 \pm 0.8 \text{ vs} 3.5 \pm 0.7$; p<0.0001). Conjunctivitis was observed in 11.3%(n=9/80); 3 patients experienced dupilumab facial redness (DFR); 17.5% (n=14/80) reported injection site reactions; 6.3% (n=5/80) discontinued treatment. 61.2 % (n=49/80) children were ineligible in the phase 3 trial.

Limitations: There is no control group. Because it was a real life study based on information from patient medical records in a French multicenter cohort, we cannot rule out the presence of reporting bias generated by the use of patient reported characteristics and missing information.

Conclusion: These real-life data confirm the efficacy and safety of dupilumab in children with moderate to severe AD extended to dyshidrosis and atopic prurigo but it also revealed a lower frequency of DFR and conjunctivitis. However, administration in injectable form may be a barrier in this age group.

Lasek Audrey (Orcid ID: 0000-0002-4105-4027)

MAZEREEUW-HAUTIER Juliette (Orcid ID: 0000-0001-6259-9790)

Chiaverini Christine (Orcid ID: 0000-0002-6063-5409)

RAISON-PEYRON Nadia (Orcid ID: 0000-0002-7991-3165)

Barbarot Sebastien (Orcid ID: 0000-0002-6629-9100)

Aubert Hle (Orcid ID: 0000-0003-1688-9900)

Droitcourt Catherine (Orcid ID: 0000-0001-6714-6643)

MAHE Emmanuel (Orcid ID: 0000-0001-5780-1827)

INTRODUCTION

Atopic dermatitis (AD) is one of the most common skin disorders in children and the leading contributor to the global burden of skin disease. (1,2) In children with moderate-to-severe AD, skin lesions often involve a large body surface area (BSA), and the related pruritus, sleep deprivation, activity restriction, reduced school performance, depression, and anxiety have a greater impact on quality-of-life (QOL) for patients and their family (3).

AD begins before the age of 5 years in more than 85% of patients and persists into adulthood in half of the cases. (4). Despite the chronic nature of AD, treatment in children is often limited to short-term topical corticosteroids (TCS), and topical calcineurin inhibitors (5-7). Although other systemic agents have been used off-label such as cyclosporine (not approved before 16 years of age in France), methotrexate, or azathioprine, the risk of serious adverse events (AE) associated with these agents and the lack of high-level evidence of long-term efficacy makes them especially difficult to manage for this age group requiring biology monitoring (8). Systemic therapies are offered only as a last resort for the most intractable cases, resulting in a large unmet need for children whose disease is inadequately controlled with topical therapy. Dupilumab (Dupixent®) is a fully human monoclonal antibody derived from VelocImmune that blocks the shared receptor component of interleukin-4 and interleukin-13. Dupilumab clinical trials have shown that these cytokines are key and central drivers of multiple type 2 inflammatory diseases. Dupilumab is approved for use in the United States and European Union and other countries for adults and adolescents with severe AD,

severe asthma with evidence of type 2 inflammation or eosinophilia, and adults with chronic rhinosinusitis and nasal polyps. Dupilumab significantly improves symptoms and quality of life in adults and adolescents with moderate-to-severe AD, with an acceptable safety profile. (9-17)

The results of a phase 3 trial of dupilumab with concomitant topical corticosteroids in children aged 6-11 years with severe AD inadequately controlled with topical therapies have been reported (11). Dupilumab was effective on symptoms and quality of life and well tolerated in children. The safety profile was consistent with that observed in adults and adolescents. Dupilumab was approved in Europe in November 2020 for children aged 6-11 years with severe AD inadequately controlled with topical therapies, and use was available by temporary authorization use (ATU). Few data are available with regard to the efficacy and safety of dupilumab in real life conditions.

This study was designed to assess the effictiveness and safety of dupilumab in children from 6 to 11 years with AD in real life, in a French multicenter retrospective cohort. We also evaluated the eligibility of our children for the published phase 3 clinical trial.

METHODS

Study design and population

In this study, we evaluated the data collected in a French multicenter retrospective cohort conducted by the research group of *Société Française de Dermatologie Pédiatrique* (SFDP) and by the *Groupe de Recherche sur l'eczéma atopique* (GREAT) of the French Society of Dermatology. Consecutive patients aged 6 to 11 years with a diagnosis of AD who received at least one injection of dupilumab between August 2018 and May 2021 were eligible for this study.

Ethical aspects

This study was approved by the CNIL (*Commission Nationale de l'Informatique et des Libertés*). Non-opposition to the use of their de-identified records was obtained for this non-interventionnal study, in accordance with French legislation.

Data Collection

A standardized questionnaire was sent to members of the GREAT and SFDP groups. The following information was collected from patient medical records: demographic variables, comorbidities, AD clinical features, previous treatments, date of introduction and dose of dupilumab, severity scores (SCORAD and investigator global assessment (IGA)) at baseline and during follow up. We evaluated the frequency and reasons of discontinuation of dupilumab at the end of follow up.

Outcomes

SCORAD and IGA were evaluated at baseline, 3 and 6 months when available. The primary outcome was the evolution of SCORAD at 3 months compared with baseline. The secondary outcome was the SCORAD at 6 months compared with baseline. We evaluated the frequency of children who achieved a 50% (SCORAD 50) or 75% (SCORAD 75) SCORAD reduction after 3 or 6 months of treatment.

We also evaluated IGA at 3 and 6 months compared with baseline and the frequency of IGA 0 or 1 (clear or nearly clear) at 3 and 6 months.

Side effects

Sides (SE) were collected. SEs were defined as the occurrences of any adverse medical condition during the treatment period. Eosinophilia was defined as a blood eosinophil count >500/mm3.

Serious SE (SSE) included SEs that resulted in death; were life-threatening; required hospitalization or prolongation of a current hospitalization; resulted in persistent or significant disability or incapacity; or required intervention to prevent permanent impairment or damage.

Assessment of eligibility

The eligibility for clinical trials for each patient was assessed retrospectively, based on criteria used in the phase 3 trial evaluating the efficacy and safety of dupilumab with concomitant topical corticosteroids in children aged 6-11 years with severe atopic dermatitis (12); "Key inclusion criteria were children age 6-11 years with AD (American Academy of Dermatology consensus criteria) diagnosed \geq 1 year before screening; Investigator's Global Assessment (IGA) score of 4, Eczema Area and Severity Index (EASI) score \geq 21, affected BSA \geq 15%, weekly averaged baseline worst itch score (Peak Pruritus Numerical Rating Scale [NRS]) \geq 4; weight \geq 15 kg; and documented history of inadequate response to topical AD medication within 6 months prior to baseline."

Statistical analysis

Quantitative data were expressed as means ± standard deviation, and qualitative data as frequency and percentages. Comparisons of means were performed using Student's t-test. A p-value below 0.05 was considered statistically significant. Statistical analyses were performed using the R software, version 3.4.3.

RESULTS

Baseline Characteristics

The study population comprised 80 patients from 21 hospitals in France. There were 46 girls (57.5%) and the mean age was 9.3 years. The characteristics of the population and disease are detailed in table 1.

Medical history

Patients had a variety of associated morbidities: atopic morbidities included asthma (n=35/80, 43,8%), allergic rhinitis (n=28/80, 35%) and food allergy (n=23/80, 28,8%).

Atopic dermatitis

The mean age of onset of AD was 12.5 months (SD 20.8); 74/80 children (92.5%) had a classic diffuse form of AD, 7.5% (n=6) had either atopic nodular prurigo (n=3) or severe palmoplantar dyshidrosis (n=3).

Baseline scorad was 53,9 \pm 18,5 and baseline IGA was 3.5 \pm 0.7; baseline Child DLQI was performed only in 10 patients with a median of 8.5; 67/71 (94.3%) patients had an IGA score of 3 to 4, and 4/71 (5.6%) had IGA score of 1 to 2. 39 children had an eosinophilia > 500 /mm3 (mean: 1457/mm3) prior to dupilumab treatment.

Treatments for AD

Accepted Articl

Systemic treatment prior to dupilumab was prescribed in 38/80 children (47.5%).

Dupilumab effectiveness: Table 2

The mean age at dupilumab initiation was 9.1 years. Mean follow up was approximately 6 months.

After 3 months of treatment, there was a significant decrease in SCORAD (median 21.8 ± 13.8 at 3

months vs. 53.9 ± 18.5 at baseline; p<0.0001) and a significant decrease in IGA (1.3 \pm 0.8 at 3 months

vs. 3.5 ± 0.7 in baseline; p<0.0001); 66.7% of children (n = 38/57) had reached an IGA of 0-1 at 3

months.

44 children of 61 (74.6%) had SCORAD50 at 3 months, 11 of 61 at 6 months (73.3%). 13/61 children

(22%) had SCORAD75 at 3 months, 4/15 at 6 months (26.6%).

AD flares

On dupilumab, 31 children experienced AD flares. The management of flares was accomplished by

the use of topical treatments: TCS or tacrolimus. Three flares were reported as head and neck

dermatitis (Figure 1).

Dupilumab safety: Table 3

10/80 children (12.5%) experienced a noninfectious ophtalmologic SE, including 9 conjunctivitis

(11,3%)(Figure 1 Severe conjunctivitis in a 9 year old boy on Dupilumab), 2 blepharitis (2.5%) and 2

ocular pruritus.

The proportion of patients with eosinophilia (>500 / mm3) was not significantly higher at 3 months

of treatment (n = 40) than before initiation (n = 39), however eosinophilia above 5000 / mm3 was

detected in 3 patients at 3 months. Fourteen (17%) children reported injection site reactions or pain.

Three children reported transient dupilumab-induced facial redness. (figure 1: Dupilumab facial

Discontinuation of dupilumab: Table 4

Five children (6,3%) stopped treatment; including 2 children who discontinued due to injection-related pain.

Eligibility: Table 5

redness)

Of the 80 patients included, 49 (61.2%), were ineligible due to at least one or more exclusion criteria for the phase 3 trial. Of these 49 patients, 33 were ineligible due to disease severity (SCORAD <50 or IGA <4) and 6 due to the clinical type of AD (atopic prurigo or palmoplantar dyshydrosis)

DISCUSSION

This retrospective observational study confirmed the real life effectiveness of dupilumab in real life and its safety in children aged 6 to 11 years with AD. The strengths of this real life study are that the patients were not selected (as they are in clinical trials) and they represented a sample of the pediatric population aged 6 and 11 year treated with dupilumab in France. Results were similar to those obtained in the clinical trials. The results of a phase 3 trial of dupilumab with concomitant TCS in children aged 6-11 years with severe AD inadequately controlled with topical the rapies have been reported (12). Dupilumab and TCS are effective and welltolerated in children with severe AD, and significantly improved quality of life. Among children treated with Dupilumab + tcs , 29.5% achieved IGA scores of 0 or 1 at 3 months. In our study, after 3 months of treatment, there was a significant decrease in SCORAD and IGA: 68.4% of children (n = 39) had reached 0-1 IGA at 3 months.

We observed a lower rate of conjunctivitis with dupilumab (11,3%) in children 6-11 years old children, compared with clinical trial results in adults (38%) (17), similar to adolescents 10.3% (14) and to children 6.7% with dupilumab q4w and 14.8% with dupilumab 100/200mg 2QW (12). We did not observe a higher proportion of patients with eosinophilia at follow up compared with baseline. However, we did observe an increase of eosinophils count in these patients. Only 3 patients had eosinophilia > 5000/mm3, leading to discontinuation of treatment. In one case, dupilumab was not effective, and in another dupilumab was effective in body symptoms but lead to facial redness. None of our patients showed clinical signs of internal organ involvement due to eosinophil infiltration. The mechanisms underlying dupilumab-induced hypereosinophilia remain unknown, but it is often associated with asthma and allergic rhinitis (17).

Dupilumab facial redness (DFR) is a side effect characterized by the paradoxical onset or worsening of facial dermatitis reported in approximately 10% of adults receiving dupilumab and is less frequent in children (18). In this study only 3 children (3%) experienced DFR. DFR may occur more frequently in post pubertal children rather than prepubertal children (35% vs 14%) with a

higher incidence with increasing age (19). The etiology of this eruption is currently unclear. One hypothesis is that DFR is a seborrheic malassezia-induced dermatitis which is consistent with the lower rate of seborrheic dermatitis in younger children. However, antifungal topics are not very effective (20).

Children with severe AD inadequately controlled with topical therapies have limited therapeutic options. Immunosuppressive treatments as cyclosporine or methotrexate do not have legal authorization (AMM) in France to treat severe AD in children; however, these molecules are used in clinical practice (8). We observed in our study an important proportion of patients who did not receive systemic treatment before dupilumab (n=42, 52.5%). In the future, the use of prior systemic treatments in severe AD in children aged 6-11 years will certainly decrease.

Another frequently reported side effect in children is pain at the injection site (n=14), which in our study led to discontinuation of treatment in 2 children. This injection site pain could be avoided by systematically prescribing a local anesthetic before injection in children. According to our experience, some children prefer using the syringe rather than the injection pen, which is more painful for them.

Phase 3 clinical trials of biotherapies are designed for a selected population. For example, in a recent real life cohort of children with psoriasis, it was shown that more than 50% of children included in the cohort were ineligible for randomized controlled trial. Thus, efficacy and safety results from phase 3 clinical trials in selected populations may not sufficiently reflect what is observed in real life, hence the need for results from real life cohort studies (22). Our study investigated whether children treated for atopic dermatitis with dupilumab in real life practice would differ from children in the original phase 3 trial from Paller (12). We found that 61.2% of patients in our cohort would not be eligible for the clinical trial. Instead of lower disease severity, IGA score and SCORAD were lower than required, probably because children had no treatment washout before starting dupilumab. The second reason was the presence of other forms than classic AD, mainly palmoplantar dyshidrosis and atopic prurigo (23,24). In daily practice in France, moderate to severe AD is characterized by SCORAD

>50 and IGA >4. (European Task Force on Atopic Dermatitis. Severity scoring of atopic dermatitis: the SCORAD index. Consensus Report of the European Task Force on Atopic Dermatitis. Dermatology 1993;186:23-31). However, these parameters have been shown to have limited relevance in real life conditions and are insufficient to determine the severity of AD. In addition, in clinical trials designed for drug approval, all previous treatments are withdrawal from patients and they don't have severe comorbidities. Pretrial wash out periods and allowing only specific "mild" medication may lead to increase in disease severity resulting in artificially high EASI and IGA scores being recorded at the beginning of the clinical trial. In the real life, patients may have severe comorbidities and have previous treatments, clinicians don't let them get worse before starting a new treatment.

Our study has some limitations. Because it was a real life study based on information from patient medical records in a French multicenter cohort, we cannot rule out the presence of reporting bias generated by the use of patient reported characteristics and missing score data for 6-month follow-up.

CONCLUSION

These real-life data confirm the effectiveness and safety of dupilumab in children with moderate to severe AD extended to dyshidrosis and atopic prurigo but also showed a lower frequency of DFR and conjunctivitis than previous studies in adolescents and adults. However, administration in injectable form may be a barrier in this age group.

References

- Mortz CG, Andersen KE, Dellgren Cel al. Atopic dermatitis from adolescence to adulthood in the TOACS cohort: prevalence, persistence and comorbidities.. Allergy. 2015; 70:836-845.
- 2. Hay RJ, Johns NE, William HC et al. The global burden of skin disease in 2010: an analysis of the prevalence and impact of skin conditions. J Invest Dermatol. 2014;134:1527-1534
- 3. Beattie PE, Lewis Jones MS. A comparative study of impairment of quality of life in children with skin disease and children with other chronic childhood diseases. Br J Dermatol 2006; 155:145-151.
- 4. E. Shaw, G.P. Currie, C.W. Koudelka, E.L. Simpson. Eczema prevalence in the United States: data from the 2003 National Survey of Children's Health, T., J Invest Dermatol, 131 (2011), pp. 67-73
- 5. A. Wollenberg, L.M. Ehmann. Long term treatment concepts and proactive therapy for atopic eczema. Ann Dermatol, 24 (2012), pp. 253-260
- 6. D. Thaçi, S. Reitamo, M.A. Gonzalez Ensenat, et al. Proactive disease management with 0.03% tacrolimus ointment for children with atopic dermatitis: results of a randomized, multicentre comparative study.BrJ Dermatol, 159 (2008), pp. 1348-1356
- 7. Wollenberg A, Barbarot S, Bieber T, Christen-Zaech S, Deleuran M, Fink-Wagner A, et al. Consensus-based European guidelines for treatment of atopic eczema (atopic dermatitis) in adults and children: part II. J Eur Acad Dermatol Venereol. 2018 Jun;32(6):850-878.
- 8. C. Flohr, A.D. Irvine. Systemic therapies for severe atopic dermatitis in children and adults. J Allergy Clin Immunol, 132 (2013), p. 774.e6
- 9. Ratchataswan T, Banzon TM, Thyssen JP, Weidinger S, Guttman-Yassky E, Phipatanakul W. Biologics for Treatment of Atopic Dermatitis: Current Status and Future Prospect. J Allergy Clin Immunol Pract. 2021 Mar;9(3):1053-1065.
- 10. Simpson EL, Paller AS, Siegfried EC, Thaçi D, Wollenberg A, Cork MJ, et al. Dupilumab

 Demonstrates Rapid and Consistent Improvement in Extent and Signs of Atopic Dermatitis

- Across All Anatomical Regions in Pediatric Patients 6 Years of Age and Older. Dermatol Ther (Heidelb). 2021 Aug 24.
- 11. Cork MJ, Thaçi D, Eichenfield LF, Arkwright PD, Sun X, Chen Z, et al. Dupilumab provides favourable long-term safety and efficacy in children aged ≥ 6 to < 12 years with uncontrolled severe atopic dermatitis: results from an open-label phase IIa study and subsequent phase III open-label extension study. Br J Dermatol. 2021 May;184(5):857-870.</p>
- 12. Paller AS, Siegfried EC, Thaçi D, Wollenberg A, Cork MJ, Arkwright PD, et al. Efficacy and safety of dupilumab with concomitant topical corticosteroids in children 6 to 11 years old with severe atopic dermatitis: A randomized, double-blinded, placebo-controlled phase 3 trial. J Am Acad Dermatol. 2020 Nov;83(5):1282-1293.
- 13. Igelman S, Kurta AO, Sheikh U, McWilliams A, Armbrecht E, Jackson Cullison SR, Kress DW, Smith A, Castelo-Soccio L, Treat J, Boothe WD, Diaz LZ, Levy ML, Patel A, Lee LW, Fraile-Alonso MC, Antaya RJ, Shah S, Kittler N, Arkin L, Siegfried E. Off-label use of dupilumab for pediatric patients with atopic dermatitis: A multicenter retrospective review. J Am Acad Dermatol. 2020 Feb;82(2):407-411.
- 14. Bansal A, Simpson EL, Paller AS, Siegfried EC, Blauvelt A, de Bruin-Weller M, Corren J, Sher L, Guttman-Yassky E, Chen Z, Daizadeh N, Kamal MA, Shumel B, Mina-Osorio P, Mannent L, Patel N, Graham NMH, Khokhar FA, Ardeleanu M. Conjunctivitis in Dupilumab Clinical Trials for Adolescents with Atopic Dermatitis or Asthma. Am J Clin Dermatol. 2021 Jan; 22(1):101-115.
- 15. Dhar S, De A, Srinivas SM. Real-World Effectiveness and Safety of Dupilumab for the Treatment of Moderate to Severe Atopic Dermatitis in Indian Patients: A Multi Centric Retrospective Study. Indian J Dermatol. 2021 May-Jun;66(3):297-301.
- 16. Simpson EL, Paller AS, Siegfried EC, Boguniewicz M, Sher L, Gooderham MJ, et al. Efficacy and Safety of Dupilumab in Adolescents With Uncontrolled Moderate to Severe Atopic Dermatitis: A Phase 3 Randomized Clinical Trial. JAMA Dermatol. 2020 Jan 1;156(1):44-56.

- 17. Faiz S, Giovannelli J, Podevin C, Jachiet M, Bouaziz JD, Reguiai Z, et al. Effectiveness and safety of dupilumab for the treatment of atopic dermatitis in a real-life French multicenter adult cohort. J Am Acad Dermatol. 2019 Jul;81(1):143-151.
- 18. Waldman RA, DeWane ME, Sloan B, Grant-Kels JM. Characterizing dupilumab facial redness: a multi-institution retrospective medical record review. J Am Acad Dermatol. 2020;82(1):230-232
- 19. Muzumdar S, Zubkov M, Waldman RA, DeWane ME, Wu R, Grant-Kels JM. Characterizing dupilumab facial redness in children and adolescents: A single-institution retrospective chart review. J Am Acad Dermatol. 2020 Nov, 83(5):1520-1521.
- 20. Darabi K, Hostetler SG, Bechtel MA, Zirwas M. The role of Malassezia in atopic dermatitis affecting the head and neck of adults. J Am Acad Dermatol. 2009;60(1):125-136.
- 21. Bax CE, Khurana MC, Treat JR, Castelo-Soccio L, Rubin AI, McMahon PJ. New-onset head and neck dermatitis in adolescent patients after dupilumab therapy for atopic dermatitis. Pediatr Dermatol. 2021 Mar;38(2):390-394.
- 22. Phan C, Beauchet A, Burztejn AC, Severino-Freire M, Mazereeuw-Hautier J, Barbarot S, et al. Evaluation of Children with Psoriasis from the BiPe Cohort: Are Patients Using Biotherapies in Real Life Eligible for Phase III Clinical Studies? Paediatr Drugs. 2019 Jun;21(3):169-175.
- 23. Giovannini M, Mori F, Oranges T, Ricci S, Barni S, Canessa C, et al. Dupilumab treatment of prurigo nodularis in an adolescent. Eur J Dermatol. 2021 Feb 1;31(1):104-106
- 24. Tilotta G, Pistone G, Caruso P, Gurreri R, Castelli E, Curiale S, et al. Our experience with prurigo nodularis treated with dupilumab. J Eur Acad Dermatol Venereol. 2021 Apr;35(4):e285-e287.

Table 1. Characteristics of the 80 children, and of the atopic dermatitis (AD)

iabie	1. Characteristics of the 80 children, and of the a	topic derinatitis (AD)				
Chil	dren					
	Gender, girls, n (%)	46 (57.5)				
	Age (y), mean ± SD	9.3 ± 2.0				
	BMI (kg.m ⁻²), mean ±SD	18.0 ± 3.7				
	Past history of HSV infection, n (%)	6 (7.5)				
Fam	Family history of atopy, n (%)					
	AD	42 (52.5)				
	Asthma	35 (43.8)				
	Rhinitis	28 (35.0)				
	Alimentary allergies	23 (28.8)				
Ato	picdermatitis					
	Age at onset (m), mean ± SD	12.5 ± 20.8				
	Main clinical type, n (%)					
	Classic AD	74 (92.5)				
	Palmoplantar dyshidrosis	3 (3.8)				
	Atopic prurigo	3 (3.8)				
	Previous eosinophilia > 500/mm³, n (%)	40 (50.0)				
	Previous general treatments for AD, n (%)					
	Cyclosporine	30 (37.5)				
	Methotrexate	21 (26.3)				
	Phototherapy	3 (3.8)				
	Omalizumab	1 (1.3)				
	Azathioprine	1 (1.3)				
	Acitretin ¹	5 (6.3)				

Dupilumab	
Age at onset (y), mean ± SD	9,1 ± 1,9
Duration of dupilumab (d), mean \pm SD 2	99.0 ± 107.2
Total cumulative years of dupilumab (y)	21.7

AD: atopic dermatitis; SD: standard deviation

¹ For atopic prurigo and palmoplantar eczema

² Include children who discontinued the treatment and those for which dupilumab is going

Table 2. Evolution of severity scores

	Basel	Baseline		M3 ± 1 month		M6 ± 1 month	
Scores	N	Value	N	Value	N	Value	
SCORAD, mean ± SD	76	53.9 ± 18.5	61	21.8 ± 13.8<0.0001	15	20.5 ± 10.0 ^{<0.0001}	
SCORAD 50, n (%)		-		44 (72.1)		11 (73.3)	
SCORAD 75, n (%)		-		13 (21.3)		4 (26.7)	
IGA, mean ± SD	71	3.5 ± 0.7	57	1.3 ± 0.8<0.0001	12	0.8 ± 0.7<0.0001	
IGA 0-1, n (%)		0		38 (66.7)		8 (66.7)	

N: number of children evaluated. IGA: investigator global assessment. SD: standard deviation Superscript: p value if <0.5, in comparison to baseline. SCORAD 50 and SCORAD 75: reduction of SCORAD compared to baseline of 50% and 75% respectively

Table 3. Adverse events in the 80 children treated by dupilumab, n (%)

AD flare	31 (38.8)
Head and neck form	3 (3.8)
Noninfectious ophtalmologic	10 (12.5)
Conjunctivitis	9 (11.3)
Blepharitis	2 (2.5)
Ocular pruritus	2 (2.5)
Eosinophilia	
> 500/mm³	40 (50.0)
> 5000/mm³	3 (3.8)
Redness head and neck	3 (3.8)
Pain or reaction injection site	14 (17.5)
Aphtous	1 (1.3)
Molluscum contagiosum	1 (1.3)
Diarrhea	1 (1.3)
Serious adverse events	0

Table 4. Discontinuation of dupilumab

Number of children, n (%)	5 (6.3)		
Duration before discontinuation (d), mean \pm SD	302.6 ± 301.4		
Causes of discontinuation, n (%)			
Adverse event	3 (60.0)		
Eosinophilia	3		
Injection pain	2		
Inefficacy	1 (20.0)		
Remission	1 (20.0)		

SD: standard deviation

Table 5. Eligibility of the 80 children in phase 3 study (12)

	Criteria not respected, N (%)
Age 6-11 years with AD	0
AD (American Academy of Dermatology consensus criteria)	6 (7.5)
AD diagnosed≥1 year	0
IGA ≥ 4	33 (41.3)
Scorad > 50 ¹	35 (43.8)
Weight≥15 kg	0
Inadequate response to topical AD medication	0
Affected BSA ≥15%,	Not evaluated
Weekly averaged baseline worst itch score ≥4;	Not evaluated
At least one criteria not respected	49 (61.3)

 $^{^{1}}$ Criteria in the clinical trial was EASI ≥ 21, that defined severe AD. Since EASI is not use in France, but SCORAD, we use the definition of severe AD, SCORAD > 50



JDV_18450_jeadv-2022-0015-File008.tiff

Figure 2. Dupilumab facial redness

